Author’s response to reviews

Title: Effects of different extracts of Curcumin on TPC1 papillary thyroid cancer cell line

Authors:
Angelica Perna (angelicaperna@gmail.com)
Antonio De Luca (antonio.deluca@unicampania.it)
Laura Adelfi (laura.adelfi@gmail.com)
Pasquale Tammaro (pasquale.tam@gmail.com)
Bruno Varriale (bruno.varriale@unicampania.it)
Teresa Esposito (teresa.esposito@unicampania.it)

Version: 3 Date: 10 Jan 2018

Author’s response to reviews:

University of Campania “L. Vanvitelli”
Department of Experimental Medicine, Molecular Genetics Laboratory, 80138 Naples Italy

Bruno Varriale
Professor
Tel+390815667621
e-mail: bruno.varriale@unicampania.it
Naples 10.01.2018

Dear Editor,

Please, find enclosed the revised copy of the manuscript “Effects of different extracts of Curcumin on TPC1 papillary thyroid cancer cell line” (BCAM-D-17-00372) for publication on "BMC Complementary & Alternative Medicine."
All scientific inaccuracies have been corrected throughout the manuscript according to the suggestions of your reviewer. All the comments have been addressed. Specific comments for the Referee are listed below.

Point 1: The authorship form has been modified as requested.

Point 2: “Conclusion” has been added before the phrase "In conclusion, these findings provide evidence that the inhibitory..." line 321.

Point 3: At line 365 the authors' contribution section has been added, as requested.

Point 4: At line 361 the statement section has been changed.

Point 5: In the Method section, line 113, the source of curcumin used has been added.

All correspondence may be addressed to myself, Bruno Varriale, Department of Experimental Medicine, Molecular Genetics Laboratory, University of Campania “Luigi Vanvitelli” Via Costantinopoli 16, 80138 Naples, Italy Tel: +390815667621; e-mail: bruno.varriale@unicampania.it

We hope that the manuscript, in its revised form, will meet all the requirements in order to be published in BMC Complementary & Alternative Medicine. Thank you in advance for your time and cooperation.

Best regards,

Prof. Bruno Varriale